Author's response to reviews

**Title:** Polyethylene glycol vs. sodium phosphate for bowel preparation: A treatment arm meta-analysis of randomized controlled trials

**Authors:**

- Ravi Juluri (rjuluri@clarian.org)
- George Eckert (geckert@iupui.edu)
- Thomas F Imperiale (timperia@iupui.edu)

**Version:** 2  **Date:** 4 June 2010

**Author's response to reviews:** see over
Dear Drs. Norton and Patel:

On behalf of my co-authors, I submit the attached manuscript entitled, “Polyethylene glycol (PEG) vs. sodium phosphate (NaP) for bowel preparation: A treatment arm meta-analysis of randomized controlled trials” for consideration as a clinical review section of the journal. This study is based on to incorporate all available evidence to determine efficacy and patient adherence to either PEG or NaP for bowel preparation prior to elective colonoscopy. To the best of our knowledge, there is no duplication of data contained within the manuscript. This paper is not under review by another journal at this time. None of the authors has any conflict of interest with the content of the manuscript.

I thank you for your consideration of our manuscript.

Sincerely yours,

Ravi Juluri, MD
Research Fellow
Division of Gastroenterology and Hepatology
Indiana University School of Medicine
Phone:(317) 630-7760, Fax: (317) 630-6611
Email: rjuluri@clarian.org